Author’s response to reviews

Title: Reporting guideline for priority setting of health research (REPRISE)

Authors:

Allison Tong (allison.tong@sydney.edu.au)
Anneliese Synnot (A.Synnot@latrobe.edu.au)
Sally Crowe (sally@crowe-associates.co.uk)
Sophie Hill (sophie.hill@latrobe.edu.au)
Andrea Matus (andrea.matusgonzalez@sydney.edu.au)
Nicole Scholes-Robertson (nicole.scholes-robertson@sydney.edu.au)
Sandy Oliver (sandy.oliver@ucl.ac.uk)
Katherine Cowan (katherine@katherinecowan.net)
Mona Nasser (mona.nasser@plymouth.ac.uk)
Soumyadeep Bhaumik (drsoomyadeepbhaumik@gmail.com)
Talia Gutman (talia.gutman@sydney.edu.au)
Amanda Baumgart (amanda.baumgart@sydney.edu.au)
Jonathan Craig (jonathan.craig@flinders.edu.au)

Version: 2 Date: 11 Dec 2019

Author’s response to reviews:

11th December 2019

Professor Tamara Hughes
Professor Eleanor Harding
Editor-in-Chief
BMC Medical Research Methodology

Re: Reporting guideline for priority setting of health research (REPRISE) BMRM-D-19-00414.R1 Response to reviewers

Dear Professor Hughes and Professor Harding

We thank the Editors and the two reviewers (Profs Sabine Van de Veer and Jacqueline Elisabeth Willy
Broerse) for their helpful comments which we have used to improve our manuscript. These are addressed as follows:

Editor comments

1. As Allison Tong is a member of the editorial board of this journal, please declare this in the Competing Interests section of the Declarations in order to maintain transparency.

As advised we have added this statement to the Competing Interests section.

Reviewer: Jacqueline Elisabeth Willy Broerse

2. I noticed one typo: on p.15, line 31: "wold" should be "would"

As noted, we have corrected this typographical error.

Kindest regards
Prof Allison Tong, on behalf of all named authors